**Author’s response to reviews**

**Title:** Annexin A1 in plasma from patients with bronchial asthma: its association with lung function

**Authors:**
Sun-Hye Lee (s2sund@gmail.com)
Pureun-Haneul Lee (puruenhanul@hanmail.net)
Byeong-Gon Kim (byeonggone@naver.com)
Hyun-Jeong Seo (svamp@naver.com)
Ae-Rin Baek (arbaek@schmc.ac.kr)
Jong-Sook Park (newstart1221@naver.com)
June-Hyuk Lee (junehyuk@schmc.ac.kr)
Sung-Woo Park (mdswpark@gmail.com)
Do-Jin Kim (kdj@schmc.ac.kr)
Choon-Sik Park (mdcspark@hanmail.net)
An Soo Jang (jas877@schmc.ac.kr)

**Version:** 3 Date: 25 Oct 2017

**Author’s response to reviews:**

Dear Editor:

Manuscript ID PULM-D-17-00094R2 entitled "Annexin A1 in plasma from patients with bronchial asthma: its association with lung function”.

We wish to express our appreciation the opportunity to revise and re-submit our manuscript. We enclose a point by point response to the reviewers’ comments. We believe the revised manuscript now fulfills the high standards of BMC Pulmonary Medicine.

With my best regards,

Correspondence
An-Soo Jang, M.D., Ph.D. Division of Respiratory and Allergy, Department of Internal Medicine, Soonchunhyang University Bucheon Hospital, 170 Jomaru-ro, Wonmi-gu, Bucheon, Gyeonggi-do, 14584, Republic of Korea.

Tel.: +82-32-621-5143
Fax.: +82-32-621-6950
E-mail: jas877@schmc.ac.kr

Editor Comments:

C1. Please list the full names, institutional addresses and email addresses for all authors on the title page.

R1. The authors added lists as you suggest the followings; Sun-Hye Lee1, Pureun-Haneul Lee1, Byeong-Gon Kim1, Hyun-Jeong Seo1, Ae-Rin Baek1, Jong-Sook Park1, June-Hyuk Lee1, Sung-Woo Park1, Do-Jin Kim1, Choon-Sik Park1, An-Soo Jang1

1Department of Internal Medicine, Soonchunhyang University Bucheon Hospital, 170 Jomaru-ro, Wonmi-gu, Bucheon, Gyeonggi-do, 14584, Republic of Korea.

Sun-Hye Lee1, s2sund@gmail.com
Pureun-Haneul Lee1, puruenhanul@hammail.net
Byeong-Gon Kim1, byeonggone@naver.com
Hyun-Jeong Seo1, svamp@naver.com
Ae-Rin Baek1, arbaek@schmc.ac.kr
Jong-Sook Park1, newstart1221@naver.com
June-Hyuk Lee1, junehyuk@schmc.ac.kr
Sung-Woo Park1, swpark@schmc.ac.kr
Choon-Sik Park1, mdcspark@hanmail.net
An-Soo Jang1, jas877@schme.ac.kr

C2. Please rename the "Introduction" section to "Background" in order for the manuscript format to meet the journal guidelines.
R2. The authors have changed as you suggest.

C3. The Availability of Data and Materials section refers to the raw data used in your study; presenting tables and figures is not sufficient to state that all data is contained within the manuscript and additional files. Please only use this statement if you have indeed provided all raw data on which your study is based. We strongly encourage all authors to share their raw data, either by providing it in a supplementary file or depositing it in a public repository and providing the details on how to access it in this section. If you do not wish to share your data, please clearly state this in this section along with a justification. For more information and a list of suitable availability statements, please see our submission guidelines: https://www.biomedcentral.com/getpublished/editorial-policies#availability+of+data+and+materials

R3. The datasets used and/or analysed during the current study are available from the corresponding author on reasonable request. The detailed clinical data is not available because of patients’ confidentiality.

C4. Please note that based on the current authors’ contributions section, several listed authors do not automatically qualify for authorship. Acquisition of funding, collection of data, writing of the manuscript or general supervision of the research group, alone, does not usually justify authorship. We ask that you ensure that the listed author contributions are in line with the ICMJE guidelines (below), and that all listed authors have performed all four points specified below.

An 'author' is generally considered to be someone who has made substantive intellectual contributions to a published study. According to the ICMJE guidelines, to qualify as an author one should have performed all 4 of the following points:

A. Made substantial contributions to conception and design, or acquisition of data, or analysis and interpretation of data;

B. Been involved in drafting the manuscript or revising it critically for important intellectual content;

C. Given final approval of the version to be published. Each author should have participated sufficiently in the work to take public responsibility for appropriate portions of the content; and

D. Agreed to be accountable for all aspects of the work in ensuring that questions related to the accuracy or integrity of any part of the work are appropriately investigated and resolved.

Please ensure author contributions listed in the authors contribution section are reflective of ICMJE guidelines. For any authorship changes, we request that a Change in Authorship form be filled out. The change in authorship form can be found from the link on the following page: https://www.biomedcentral.com/submissions/editorial-policies#authorship
All instructions can be found on the form; please treat the 'current authorship' section as the original authorship. Please return the form within 14 days with all author signatures (including those newly added/removed). Please also ensure that the relevant authors change the statements in column 4 of the signature table to reflect whether their name is being added/removed.

R4. We added author’s contributions

SHL was performed the experiments, drafted the manuscript. PHL and BGK performed the statistical analysis and ELISA. HJS, ARB, JSP, JHL, SWP, DJK, CSP were involved in the acquisition of the data. ASJ was involved in the concept and design of the study and helped to draft the manuscript, interpretation of the data and participated. All authors read and approved the final manuscript.

C5. In the Funding section, please indicate the role of the funding body in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript.

R5. We indicate funding source the following.

This research was supported by a grant of the Korea Health Technology R&D Project through the Korea Health Industry Development Institute (KHIDI), funded by the Ministry of Health & Welfare, Republic of Korea (grant number: HI15C2032) and Soonchunhyang University Research Fund.

C6. Please include the accession numbers for the sequence data generated in your study in the Availability of Data and Materials section of the manuscript.

R6. The authors confirmed change of anxa1 protein expression in asthma, doesn’t exist the accession numbers for the sequence dataset.

Availability of data and materials

The datasets used and/or analysed during the current study are available from the corresponding author on reasonable request. The detailed clinical data is not available because of patients’ confidentiality.

C7. Please re-write text overlap in the Background and Discussion sections. This text overlaps with text in other, previously published, papers. The iThenticate report has been attached for your convenience.

R7. The authors re-write text in the background and discussion sections.